Author's response to reviews

Title: A Rare Variant of Superficial Ulnar Artery, and its clinical implications: Case Report

Authors:

  Kithsiri J Senanayake Dr (kithsirijanakantha@yahoo.com)
  Sujee Salgado Dr (sujeesal@yahoo.com)
  Manjula Jayanthi Dr (manjulajayanthi@yahoo.com)
  Ranil Fernando Dr (ranilfern@sltnet.lk)
  Kosala Somarathne Dr (kosalasomarathne@yahoo.com)

Version: 2 Date: 12 July 2007

Author's response to reviews:

Dear Sir/ Madam,

following changes were done to the manuscript in revision

1. statement regarding competing interests
2. acknoladement
3. statement regarding the consent
4. figure legends add to the bottom of the reference list

thank you